Author's response to reviews

Title: Porencephaly and psychosis: a case report and review of the literature

Authors:

Athanassios Douzenis (thandouz@med.uoa.gr)
Emanouil N Rizos (erizos@med.uoa.gr)
Athanassia Papadopoulou (ilias78@otenet.gr)
Matilda Papanassioti (matildapapathanasiou@hotmail.com)
Lefteris Lykouras (panpsyclin@attikonhospital.gr)

Version: 5 Date: 24 February 2010

Author's response to reviews:

Dear Editor

Following your instructions we have made all major revisions suggested (names of authors, affiliations etc). We have provided a title for table 1, we have included the figure legends in the manuscript. We have given each figure its own title and legend.

Regarding the minor revisions we did not remove the key words (did not know where to place them). We did not remove the vertical lines. All other changes were made.

Yours sincerely

A. Douzenis